# NBER WORKING PAPER SERIES

# CORPORATE TAXATION AND THE EFFICIENCY GAINS OF THE 1986 TAX REFORM ACT

Jane G. Gravelle

Laurence J. Kotlikoff

Working Paper No. 3142

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 October 1989

The opinions expressed in this paper are those of the authors and do not necessarily reflect the views of The Congressional Research Service of the Library of Congress, Boston University, or the National Bureau of Economic Research. We are particularly grateful to Alan Auerbach and Oldrich Kyn for very extensive and critical discussions. We also thank Don Fullerton, Yolanda Henderson, and Tom Woodward for their very helpful comments. This paper is part of NBER's research program in Taxation.

NBER Working Paper #3142 October 1989

CORPORATE TAXATION AND THE EFFICIENCY GAINS OF THE 1986 TAX REFORM ACT

#### ABSTRACT

The 1986 Tax Reform Act, while having little effect on the overall effective tax rate on U.S. capital income, did reduce significantly the difference in effective taxation of corporate and noncorporate capital within a number of U.S. industries. The Mutual Production Model developed in Gravelle and Kotlikoff (1989) can be used to study the efficiency gains from the reduction in corporate tax wedges within industries. Unlike the Harberger Model, the Mutual Production Model permits both corporate and noncorporate firms to produce the same goods and, therefore, to coexist within a given industry.

This paper develops an 11 industry - 55 year dynamic life cycle version of the Mutual Production Model. We use this model to study the steady state efficiency gains associated with the new law. While we do not simulate the economy's transition path, our steady state welfare changes are those that arise from compensating transitional generations for the first order redistribution of income associated with the Tax Reform.

We find that the 1986 Tax Reform law reduces excess burden by .85 percent of our model's economy's present value of consumption. This efficiency gain reflects the Tax Reform's reduction in corporate non-corporate tax wedges, particularly in those industries with significant non-corporate production. Measured as a flow the 1988 estimated efficiency gain from the Tax Reform Act is \$31 billion.

Jane G. Gravelle Congressional Research Service Room LM-213, Madison Building Library of Congress 10 First Street, SE Washington, DC 20024-3615 Laurence J. Kotlikoff Boston University and National Bureau of Economic Research 1050 Massachusetts Avenue Cambridge, MA 02138 The 1986 Tax Reform Act represented a sweeping change in tax law. While the dramatic reductions in personal tax rates have received most of the attention, the new law also greatly altered corporate tax wedges. In some industries, such as agriculture, the new tax law reduced by over one quarter the difference between marginal taxation of corporate and noncorporate capital income. In other industries, such as mining, the corporate tax wedge actually increased.

Conventional wisdom, that relies on various incarnations of the Harberger (1962, 1964, and 1966) Model, suggests rather small efficiency effects from these changes in corporate tax wedges. But the Harberger model appears to understate greatly the efficiency effects of changes in the corporate tax wedge for the following reason: the Harberger model does not admit production of the same good or collection of goods by both corporate and noncorporate firms. In empirical applications of the Harberger model the fact that both corporate and noncorporate production occurs in the same industry is finessed by treating all firms within an industry as identical firms facing the industry—wide average rate of taxation. This procedure misses entirely differences within industries in the tax treatment of corporate and noncorporate firms.

In Gravelle and Kotlikoff (1989, 1988) we present two models in which corporate and noncorporate firms coexist within the same industry. These two 2 sector models have different structures, but their common characteristic - a very high infinite within-industry elasticity of substitution between corporate and noncorporate output - implies an excess burden from corporate taxation that is many times larger than that predicted by the Harberger Model. The new models suggest that the efficiency gains from changes in corporate tax

wedges in the 1986 Tax Reform Act may be much larger than those previously estimated (see Fullerton, Henderson, and Mackie, 1987).

This paper develops an 11 sector 55 year dynamic life cycle version of the Gravelle-Kotlikoff (1989) model and uses this model to study the efficiency effects of changes in corporate tax wedges under the new law. Our model suggests that the 1986 Tax Reform Act reduced the excess burden of the U.S. tax structure by .85 percent of the present value of consumption. On an annual basis this represents \$31 billion based on the 1988 level of U.S. consumption.

This paper's model features mutual production of the same good (collection of goods) by corporate and noncorporate firms in 8 of 11 industries. In these industries there is an upward sloping supply of noncorporate output reflecting the less than perfectly elastic supply of entrepreneurial talent available for running noncorporate enterprises. Increases in corporate tax wedges within these industries lead existing proprietors to expand their output and lead less able entrepreneurs to establish proprietorships in these industries. But non-corporate production does not drive out corporate production. While they are at a tax disadvantage, corporations are still able to compete with the less able entrepreneurs. A second feature of the model, namely the requirement that corporate firms produce above a minimum scale, ensures that corporate production cannot be repackaged into small proprietorships and thereby avoid the corporate tax. These two features notwithstanding, corporate and noncorporate firms in each of the 8 mutual production industries produce with the same constant returns to scale production function.

In addition to permitting mutual production of the same good by corporate and noncorporate firms, the new model has a number of advantages compared to

our previous models and, in many cases, compared to the literature. First, the model merges statics with dynamics through a 55 year life cycle model of intertemporal choice. Second, it uses a fixed coefficient input-output table to take account of intermediate inputs. Third it deals explicitly with depreciation, a factor that can alter efficiency calculations. Fourth, it takes account of personal as well as corporate marginal taxation of capital income. Fifth, it takes account of differences across industries in marginal corporate tax wedges. Sixth, it models owner-occupied housing as a separate industry whose sole input is noncorporate capital. And seventh, it takes account of the use and production of different types of capital goods.

In any intertemporal model of this kind one must be concerned not to confuse intergenerational redistribution with economic efficiency. In this paper we develop a variant of Auerbach's (1988) method of separating efficiency from redistribution. Our procedure entails compensating, in a lump sum manner, each generation (initial, transitional, and new steady state generations) for the first order income changes they experience as a consequence of the intergenerational incidence of policy changes. In addition, our compensation scheme has the feature that the efficiency gains from the tax reform are distributed to initial, transitional, and final steady state generations in proportion to their initial consumption. 2 This method of compensation has two advantages. First, it appears to mitigate differences in welfare changes between transitional and final steady state generations, which is important given that we only simulate steady states in this paper. Second, ignoring transitional differences in welfare changes, the change in welfare of a representative individual alive in the new steady state measured as a fraction of the present value of his lifetime consumption equals the ratio of the present value of the economy's total (transitional plus new steady state)

efficiency gain from tax reform to the present value of the economy's total (transitional plus new steady state) consumption. Thus, ignoring transitional differences, the percentage change in individual welfare also equals, and thus provides the correct measure of, the aggregate economy's efficiency gain.

The paper proceeds in the next section, II, with a description of the supply equations of the model. Section III presents the demand side of the model. Section IV provides the equilibrium conditions. Section V discusses the compensation scheme. Section VI describes the model's calibration and method of solution. Section VII presents results, and Section VIII summarizes and concludes the paper.

# II. The Supply Side of the Model

# A. Corporate Production

The model has three factors of production: capital, labor, and managerial input (entrepreneurial input in the case of noncorporate firms). Each agent chooses whether to be a corporate manager, a worker, or an entrepreneur. There are 11 industries/sectors corresponding (in order 1 to 11) to rental housing, agriculture, mining, oil and gas, construction, transportation, services, trade, manufacturing, utilities, and owner-occupied housing. We discuss our modeling of owner-occupied housing, indexed as industry 11, below. The manufacturing and utility industries are totally corporate. The production functions for corporate firms in industries 1 to 10 are given in equation (1):

In (1)  $Q_{ci}$  stands for corporate output in sector i. This output is determined by a production function that is fixed coefficients in a value added

production function,  $F_{ci}$ , and the level of intermediate inputs indexed by  $V_{ci}$ . The terms  $\theta_i$  and  $\theta_i$  are coefficients. The value added production function is given by:

(2) 
$$\mathbf{F}_{ci} = \left\{ \begin{array}{ccc} \mathbf{f}_{ci}(D_i \mathbf{M}_{ci}, \mathbf{L}_{ci}, \mathbf{K}_{ci}) & \mathbf{M}_{ci} \ge \underline{\mathbf{M}}_{ci} \\ 0 & \mathbf{M}_{ci} < \underline{\mathbf{M}}_{ci} \end{array} \right\}$$
  $i=1,10$ 

In (2)  $D_1$  is the managerial efficiency parameter in sector i.  $M_{c1}$  stands for corporate managers in sector i,  $L_{c1}$  stands for corporate labor in sector i, and  $K_{c1}$  stands for corporate capital in sector i. Corporate firms can only produce if they produce above a minimum scale which we model in terms of a minimum number of managers,  $\underline{M}_{c1}$ . The capital input,  $K_{c1}$ , represents a composite of three capital goods - structures\land, equipment, and inventories - each of which is used in fixed proportions. The coefficients determining these proportions are, however, industry-specific.  $K_{c1}$  is defined by:

(3) 
$$K_{ci} = \min (1/\psi_i S_{ci}, 1/\xi_i E_{ci}, 1/\omega_i I_{ci})$$
 i=1,10

where  $S_{ci}$ ,  $E_{ci}$ ,  $I_{ci}$  are, respectively, structures/land, equipment, and inventories used in corporate production in sector i. The terms  $\psi_i$ ,  $\xi_i$ , and  $\omega_i$  are the industry-specific coefficients determining the capital composite.

The intermediate input used in the production of  $Q_{c1}$  also represents a fixed coefficients composite with industry-specific coefficients. Equation (4) defines the intermediate input:

(4) 
$$V_{ci} = \min (1/v_{1i}Q_{1ic}^{v}, 1/v_{2i}Q_{2ic}^{v}, \dots, 1/v_{10i}Q_{10ic}^{v})$$
  $i=1,10$ 

In (4) the terms  $v_{ji}$  (j=1,10) are input-output coefficients, and the terms  $Q_{jic}^{v}$  (j=1,10) are the intermediate inputs of good j used in sector i by corporate firms.

Since the production relation (1) satisfies constant returns to scale, one can express corporate output per manager in industry i,  $q_{ci}$ , as well as intermediate input per manager,  $v_{ci}$ , in terms of corporate labor per manager,  $l_{ci}$ , and corporate capital per manager,  $k_{ci}$ , as in (5):

# B. Noncorporate Production

Each agent in the economy can potentially become an entrepreneur in one of the 8 industries (indexed 1 to 8) that have noncorporate production; alternatively, the agent can be a manager or a worker. While all agents are equally productive as managers or workers, as entrepreneurs their productivity depends on their efficiency coefficients. The h(,,,) function in equation (5) also governs noncorporate output per entrepreneur,  $q_{ni}$ , and intermediate input per entrepreneur,  $v_{ni}$ , in industry i, but the coefficient  $D_i$  is replaced by  $A_i$ . Equation (5) expresses the output of an entrepreneur in industry i whose industry i efficiency coefficient is  $A_i$ .

(6) 
$$q_{ni}(A_i) = 1/\theta_i v_{ni} = 1/\theta_i h(A_i, l_{ni}, k_{ni})$$
 i=1,8

The terms  $\mathbf{1}_{ni}$  and  $\mathbf{k}_{ni}$  stand, respectively, for the amounts of labor and capital hired by the entrepreneur. As in the case of corporate production, noncorporate capital used in industry i is a composite of the three types of capital with the composition of this composite determined by (3).

In equilibrium only those entrepreneurs of high ability will set up shop. Let  $\underline{A}_1$  index the least able entrepreneurs to set up shop in industry i. Agents with a coefficient  $\underline{A}_1$  that is less than  $\underline{A}_1$  can earn more as a worker or manager than as an entrepreneur in industry i. Total noncorporate output in industry i,  $\underline{Q}_{ni}$ , is given by:

(7) 
$$Q_{ni} = \int_{\underline{A}_{1}}^{\overline{A}_{1}} \int_{0}^{\underline{A}_{1}} q_{ni}(\underline{A}_{1}) z(\underline{A}_{1}, \underline{A}_{2}, \dots, \underline{A}_{8}) d\underline{A}_{1} d\underline{A}_{2} \dots d\underline{A}_{8}$$
  $i=1,8$ 

In (7) the term  $\overline{A}_1$  is the maximum efficiency coefficient. As described below, entrepreneurial profits in industry i are proportional to  $A_1$ . Equation (7) adds up the output of all industry i entrepreneurs. Each of these entrepreneurs has an entrepreneurial efficiency coefficient for industry i,  $A_1$ , that satisfies two requirements. First,  $A_1$  must at least equal the minimum value  $\underline{A}_1$ , i.e., the agents must prefer to set up shop in industry i than become a worker and manager or at least be indifferent between these two choices. Second, the agent must prefer to be an entrepreneur in industry i than in each of the other industries  $j\neq i$  (i=1,8). In (7) the 7 integrals after the product sign check that the agent can't earn more as an entrepreneur in industry  $j\neq i$  than he can as an entrepreneur in industry i (i=1,8).

To understand the limits of these integrals first note that an agent with efficiency coefficient  $\underline{\mathbb{A}}_j$  in industry j is just indifferent between being an entrepreneur in industry j or a worker or a manager. Next recall that entrepreneurial profits are proportional to the entrepreneurial efficiency coefficients. Hence an agent whose industry i coefficient is  $\underline{\mathbb{A}}_i$  earns  $\underline{\mathbb{A}}_i/\underline{\mathbb{A}}_i$  times the earnings of a worker or manager. For this agent to prefer to be an entrepreneur in industry i rather than j, his industry j coefficient must be less than  $(\underline{\mathbb{A}}_i/\underline{\mathbb{A}}_i)\underline{\mathbb{A}}_j$ . For example, if  $\underline{\mathbb{A}}_i/\underline{\mathbb{A}}_i$  is 3 the agent could make more as an entrepreneur in industry j than as an entrepreneur in industry if his  $\underline{\mathbb{A}}_j$  exceeded 3 times  $\underline{\mathbb{A}}_j$ , which is the industry j coefficient of an entrepreneur who is just indifferent between being a worker or manager.

# C. Factor Prices

To determine the pre-tax rental rate of capital in a particular industry, we need first to clarify the market price of each industry's composite capital good. Recall that the capital composites consist of structures\land, equipment, and inventories. In the model structures\land are produced by industry 5, the construction industry; hence, the price of structures\land is denoted  $P_5$ . Equipment as well as non-equipment manufactures are modeled as a single good produced by the manufacturing industry, industry 9. The price of equipment is, therefore, denoted  $P_9$ . Inventories of good i sell for price  $P_i$ . With these three prices we can express the price,  $P_{ki}$ , of a unit of capital used in industry i as:

The pre-tax rental on a unit of corporate capital used in industry i is denoted by  $R_{\text{ci}}$ . The pre-tax rental on a unit of noncorporate capital in industry i is  $R_{\text{ni}}$ . These two rentals are given by:

where r is the net rate of return,  $\tau_{\text{ci}}$  is the corporate tax levied (for ease of exposition and without loss of generality) on the net return in industry i,  $\tau_{\text{ni}}$  is the tax on the net return to noncorporate capital in industry i, and  $\Delta_{\text{ki}}$  is the value of depreciation on a unit of capital in industry i. The tax rate  $\tau_{\text{ci}}$  is a marginal effective tax rate on new investment. In the empirical implementation of the model we derive the effective tax rate from a discounted cash flow model. The discounted cash flow model accounts for the deductibility of interest, the deferral of taxes through accelerated

depreciation, the taxation of capital gains on a realization rather than an accrual basis, and investment incentives including the investment tax credit. Because some of these tax features differ by industry the  $\tau_{\rm ci}$ 's and  $\tau_{\rm ni}$ 's will vary across industries.

Since inventories do not depreciate,  $\Delta_{\mathbf{k}\,\mathbf{i}}$  is given by:

(11) 
$$\Delta_{ki} = \delta_{i5} \psi_i P_5 + \delta_{i9} \xi_i P_9$$
 i=1,10

where  $\delta_{15}$  and  $\delta_{19}$  are the respective depreciation rates of structures/land and equipment in industry i.

The other two factors of production, workers and managers, have an identical factor price denoted by the wage, W.<sup>4</sup> This reflects our assumption that agents are equally productive either as workers or managers; since agents who don't choose to become entrepreneurs are free to choose to be either workers or managers, they must, in equilibrium, receive the same compensation.

#### D. Profit Maximization and Factor Demands

In maximizing profits both corporate and noncorporate firms set the marginal revenue product of factors equal to their pre-tax rentals. The solution to these conditions permits us to write the industry i corporate demands for capital per manager,  $k_{ci}$ , and labor per manager,  $l_{ci}$ , as functions of the managerial efficiency coefficients and the wage and the pre-tax capital rental:

(13) 
$$l_{ci} = D_i b_i (W/P_i, R_{ci}/P_i)$$
  $i=1,10$ 

The corresponding demands for capital per entrepreneur and labor per entrepreneur are:

$$k_{ni} = A_i e_i (W/P_i, R_{ni}/P_i)$$
 i=1,8

The first order condition for hiring managers together with equations (12) and (13) permit writing profits per corporate manager in sector i,  $\pi_{ci}$ , as a function of W,  $R_{ci}$ , and  $P_i$ . In equilibrium profits per corporate manager are paid out to the corporate managers as their wage, hence we have:

(16) 
$$\pi_{ci} = D_i x_i (W, R_{ci}, P_i) = W$$
  $i=1,10$ 

Profits of an entrepreneur in sector i with efficiency coefficient  $A_1$ ,  $\pi_{ni}(A_1)$ , can be written using the  $x_i(\cdot,\cdot)$  function as:

(17) 
$$\pi_{ni}(A_i) = A_i \pi_i(W, R_{ni}, P_i)$$
 i=1,8

# E. Choice of Occupation

In deciding whether to be an entrepreneur or to be a worker or manager, each agent considers the profits he would make as an entrepreneur in industry (i=1,8) as well as the wage paid to workers and managers. The marginal entrepreneur in industry i is just indifferent between being an entrepreneur and being a worker or manager. Recall that  $\underline{A}_{i}$  (i=1,8) denotes the efficiency coefficient of the marginal entrepreneur in industry i (1=1,8); hence,  $\underline{A}_{i}$  (i=1,8) satisfies:

(18) 
$$\pi_{ni}(\underline{A}_i) = \underline{A}_i x_i (W, R_{ni}, P_i) = W$$
 i=1,8

Combining (17) and (18) implies:

(19) 
$$\pi_{ni}(A_i) = (A_i/A_i)W$$
 i=1,8

From (19) agents who are just indifferent between being entrepreneurs in industry j (j=1,8) and industry k (k=1,8) satisfy:

(20) 
$$\pi_{nj}(A_j) = \pi_{nk}(A_k)$$
 or  $A_j/A_j = A_k/A_k$  j=1,8, k=1,8

Consideration of equation (20) further clarifies the limits of integration in (7).

#### F. Owner-Occupied Housing

Owner-occupied housing, designated as industry 11, differs from the other 10 industries in that value added of housing services depends only on the amount of capital used in owner-occupied housing. As stated in equation (21), value added in the flow of owner-occupied housing services,  $F_{11}$ , is assumed to be proportional to the stock of capital used in owner-occupied housing,  $K_{11}$ , which consists of only structures/land.

$$(21) F_{11} = H_{11}K_{11}$$

The pre-tax rental on housing, R<sub>11</sub>, is given by:

(22) 
$$R = rP_{k11}(1+\tau_{11}) + \Delta_{11}$$

#### III. The Demand Side

In the actual simulation of the model we use a 55 year (age 21 through age 75) continuous time utility function that is additively separable over time and CES over goods at a point in time. However, for ease of notation and exposition we present the remainder of the model using a two period model; the 2 period CES utility function is given by:

(23) 
$$U_{t} = \frac{1}{(1-1/\eta)} \left[ g_{1} c_{1,1,t}^{1-1/\varphi} + \dots + g_{11} c_{11,1,t}^{1-1/\varphi} \right]^{(1-1/\eta)/(1-1/\varphi)}$$

$$+ \frac{1}{(1-1/\eta)(1+\rho)^{1/\eta}} \left[ g_{1} c_{1,2,t+1}^{1-1/\varphi} + \dots + g_{11} c_{11,2,t+1}^{1-1/\varphi} \right]^{(1-1/\eta)/(1-1/\varphi)}$$

In the equation  $c_{k,j,t}$  stands for consumption of good k at age j at time t, the term  $\eta$  is the intertemporal elasticity of substitution, the term  $\varphi$  is the intratemporal elasticity of substitution, and  $\rho$  is the rate of time preference. We raise the expression  $(1+\rho)$  to the power  $1/\eta$  because in some of our simulations, in which we consider a zero intertemporal elasticity of substitution, we still want to have a rising age-consumption profile. Blowing up the discount factor by  $1/\eta$  produces that result.

Equation (24) expresses the steady state lifetime budget constraint for a worker or manager. The budget constraint for entrepreneurs is identical except that we must substitute the entrepreneur's profit for the wage.

(24) 
$$\sum_{j=1}^{2} \sum_{k=1}^{11} \frac{P_k c_{j,k}}{(1+r^g)^{j-1}} = W - m + \frac{m(1+n)}{1+r^g}$$

The first term on the right hand side of (24) is first period labor earnings, W. In the second period we assume the individual is retired. (In the empirical analysis we assume 40 years of full time labor supply.) The m's reflect the compensation scheme described in the next section. The parameter n is the population plus productivity growth rate. The left hand side of (24) gives the present value of expenditure on the 11 goods. The interest factor rg stands for a weighted average of gross of tax rates of return on the model's capital goods, where the weights are the shares of the value of each capital asset in the total value of capital assets, E. More formally:

(25) 
$$\mathbf{r}^{\mathbf{g}} = \sum_{i=1}^{10} s_{ci} (R_{ci}^{-\Delta_{i}}) / P_{ki} + \sum_{i=1}^{8} s_{ni} (R_{ni}^{-\Delta_{i}}) / P_{ki}^{+} + s_{11} (R_{11}^{-\Delta_{11}}) / P_{k11}^{-}$$

where:

(26) 
$$s_{ci}^{-} = \frac{P_{ki}K_{ci}}{E}$$
,  $s_{ni}^{-} = \frac{P_{ki}K_{ni}}{E}$ , and  $s_{11}^{-} = \frac{P_{k11}K_{11}}{E}$ 

and

(27) 
$$E = \sum_{i=1}^{10} P_{ki} K_{ci} + \sum_{i=1}^{8} P_{ki} K_{ni} + P_{ki1} K_{11}$$

Equation (24) is precisely the same budget constraint that arises if future consumption and labor earnings are discounted at the net rate of return, but capital income and wage taxes are rebated in a lump sum manner to each agent at the time the capital income taxes are paid. Hence, the budget constraint (24) incorporates the standard assumption in studies of this kind, namely that the government returns tax payments to the consumers in a lump sum manner.

# IV. Equilibrium

Table 1 presents the equations determining the steady state equilibrium of the model. All quantities are measured in efficiency units, i.e., they are adjusted for population and productivity growth. Equation (28) states that the demand for labor must equal the supply of labor. In the equation  $\bar{L}$  stands for the number of agents. The first term on the left of (28) gives the demand for managers and workers by corporations, while the second term on the left of (28) counts the number of entrepreneurs and employees of entrepreneurs. Equation (29) simply defines aggregate capital within industry i as  $K_{\bar{1}}$ . In equation (29) the terms  $k_{n\bar{1}}$  for i=9,10 equal zero. For i=11  $K_{\bar{1}}$  is simply aggregate owner—occupied housing capital.

Equation (30) defines aggregate output in sector i,  $Q_1$ , as the sum of corporate and noncorporate output. For i=11  $Q_1$  is total owner-occupied housing output. Equation (31) defines the total demand for each commodity as

the sum of the demands of each age group. Equation (32) states that the supply of output in sectors other than 5 and 9 must equal the sum of consumption demand, the demand for intermediate inputs, and the inventory investment demand. In the steady state inventory investment must equal n, the rate of population plus productivity growth, times the stock of inventories. Equations (33) and (34) are equilibrium conditions (supply equals demand) for, industry 5, construction, and industry 9, manufacturing. In addition to consumption, intermediate input, and inventory demand, there is a demand for new construction to add to the stock of structures used in different industries to accommodate population and productivity growth as well as to replace structures in different industries that have depreciated. In the case of manufacturing there is a demand to add to the stock of equipment (produced by the manufacturing sector) at the rate of population plus productivity growth as well as a demand to replace depreciating equipment.

Equation (35) relates the price of each good to its factor input and intermediate goods prices. Note that for the price of housing services (sector 11) the wage rate does not appear since the stock of housing capital is the only input in housing services. Equation (36) is a rewrite of (18) and is used to determine minimum entrepreneurial abilities in each industry. Equations (37) and (38) are the first order conditions for utility maximization. Equation (39) is the budget constraint for a new worker or manager; for an entrepreneur wage earnings is replaced by the entrepreneur's profits. Finally, equation (40) indicates that we have taken aggregate consumption expenditure as our numeraire.

We next ask whether the number of unknowns equal the number of equations in Table 1. In doing so let us ignore the purely definitional variables  $K_i$  (i=1,10),  $Q_i$  (i=1,10), and  $C_i$  (i=1,11) and the equally numerous equations (29)

through (31). Let us also use equations (37) and (38) to write all  $c_{ji}$ 's in terms of  $c_{11}$ . Table 1 then contains 32 variables. These variables are 10  $M_{ci}$ s,  $K_{11}$ , 8  $\underline{A}_{i}$ s, 10  $P_{i}$ s, 1 W, 1 r (recall that  $l_{ci}$ ,  $l_{ni}$ ,  $k_{ci}$ , and  $k_{ni}$  depend on W and r), and  $c_{11}$ . There are also 32 equations. These equations are 1 in equation (28), 11 in equations (32) through (34), 10 in equation (35), 8 in equation (36), 1 in equation (40), and 1 in equation (41).

# V. Separating Efficiency from Redistribution

In any dynamic analysis of excess burden it is critically important to consider the welfare of generations living during the transition between steady states as well as those living in the policy-induced new steady state. Policy changes may make those living in the new steady state better off (worse off), not because such policies are more efficient, but because they make earlier generations worse off (better off). Such redistribution is simply the intergenerational expression of the incidence of the fiscal policy in question. Intergenerational incidence refers here to changes in the real tax burdens of different generations, where real tax burdens may differ from nominal tax burdens due to induced general equilibrium changes in factor and commodity prices along the economy's transition path.

In static analyses the standard way to distinguish efficiency gains from welfare changes due to redistribution is to restore, in a lump sum manner, the changes in real income associated with the policy's incidence (but not its excess burden). Changes in welfare remaining after this first order restoration of real income correspond simply to pure efficiency changes.

Auerbach (1988) suggested this strategy for analyzing the steady state pure efficiency gain in dynamic models. His suggestion involves compensating, in a lump sum manner, those living in the new steady state not only for the

differences in nominal taxes paid between the new and the previous steady state, but also for differences in factor rewards and commodity prices between the new and the previous steady state. While Auerbach focuses only on the steady state, to reach his compensated steady state one must also compensate each transitional generation for any first order income changes it experiences due to the policy's intergenerational incidence. These generation—specific compensation payments sum to zero in present value. Given this compensation scheme the present value sum of the dollar measure of each generation's change in utility equals the policy's total excess burden.

To see these points consider first the change in utility of the generation that is old at time t when the policy is changed by a small amount. Equation (43) expresses the change in utility for a representative member of this generation:

(42) 
$$\frac{dU_{t-1}}{\lambda_{t-1}} = \sum_{i=1}^{11} P_i dc_{i2t}$$

In (42) the utility of the old at time t is denoted  $U_{t-1}$  since the old at time t were born at t-1. The term  $\lambda_{t-1}$  stands for the marginal utility of income of the old at time t. Again the subscript t-1 is used to denote the time at which the generation was born.

Equation (43) expresses the compensated budget constraint of an old individual at time s. The term  $m_s$  ( $s \ge t$ ) stands for the lump sum compensation payment received by an old individual at time s. The compensation scheme is designed such that what the old generation receives in a given period equals what the young give up in that period. Since the government's net compensation payments to different generations sum to zero in each period, they clearly sum to zero in present value.

For expositional ease we ignore depreciation in equation (43) and the rest of this section. The terms  $P_{is}$ ,  $P_{kis}$ , and  $k_{is}$  in equation (43) stand, respectively, for the price of output i in period s, the price of capital in industry i in period s, and total capital in industry i per old person in period s. Equation (43) states that the expenditure of the elderly at time s equals the value of their assets plus their pre-tax asset income (the second summation in the right hand side of (43)), plus the value of the lump sum compensation,  $(1+n)m_s$ . The differential of the term  $m_s$  is defined in equation (44).

$$\frac{11}{dm_{s}} = \sum_{i=1}^{L} dP_{i}c_{12} - \sum_{i=1}^{L} dP_{kis}k_{is} - d \begin{bmatrix} 11 \\ r_{s} \sum_{i=1}^{L} P_{kis}(r_{cis}k_{cis} + r_{nis}k_{nis}) \end{bmatrix} - r_{s} \sum_{i=1}^{L} P_{kis}k_{is}k_{is} - d \begin{bmatrix} 11 \\ r_{s} \sum_{i=1}^{L} P_{kis}(r_{cis}k_{cis} + r_{nis}k_{nis}) \end{bmatrix} - r_{s} \sum_{i=1}^{L} P_{kis}k_{is}k_{is} - d \begin{bmatrix} 11 \\ r_{s} \sum_{i=1}^{L} P_{kis}(r_{cis}k_{cis} + r_{nis}k_{nis}) \end{bmatrix} - r_{s} \sum_{i=1}^{L} P_{kis}k_{is}k_{is} - d \begin{bmatrix} 11 \\ r_{s} \sum_{i=1}^{L} P_{kis}(r_{cis}k_{cis} + r_{nis}k_{nis}) \end{bmatrix} - r_{s} \sum_{i=1}^{L} P_{kis}(r_{cis}k_{cis} + r_{nis}k_{nis}) \end{bmatrix} - r_{s} \sum_{i=1}^{L} P_{kis}(r_{cis}k_{cis} + r_{nis}k_{nis})$$

where  $k^{C}_{is}$  and  $k^{n}_{is}$  stand for corporate and noncorporate capital per old person in sector i in period s, and  $\alpha$  is the share of aggregate consumption consumed by the old. The term  $\alpha=1/[1+(1+\rho)(1+n)/(1+r_{s})^{\eta})$ . The term  $\mathbf{x}_{is}$  stands for net output of good i at time s per old person.

Together, the differential of equation (43) and (44) imply:

The term in brackets on the right hand side of (45) is the period s change in excess burden arising from a small change in the tax structure. Hence, according to equations (42) and (45) and the fact that dk<sub>it</sub> equals zero (since the initial stocks of capital are given) the compensation to each initial (time t) elderly individual is such that his change in utility from a small change in the tax system equals the economy's change in excess burden in the period he is old multiplied by his share of consumption; i.e., the old and, as shown below, the young share each period's change in excess burden in proportion to their share of aggregate consumption.

The change in lifetime utility of a young member of generation s  $(s \ge t)$  is given by:

(46) 
$$\frac{dU_{s}}{\lambda_{s}} = \sum_{i=1}^{L} P_{is} dc_{i1s} + \frac{1}{1+r_{s}} \sum_{i=1}^{L} P_{is} dc_{i2s}$$

To determine the value of the right hand side of (46) we consider the division of labor income of the young at time s between their consumption and their acquisition of assets:

(47) 
$$\sum_{i=1}^{11} P_{is} c_{i1s} + \sum_{i=1}^{11} P_{kis} k_{is+1} = W_s - m_s$$

The substitution of  $dm_s$  from (44) into the differential of (47) for  $s \ge t$  implies, together with equation (45):

(48)

$$\frac{11}{\sum_{i=1}^{\Sigma P} i s} dc_{i1s} + \frac{1}{1+r_s} \frac{11}{\sum_{i=1}^{\Sigma} P_{is}} dc_{i2s} = \underbrace{\frac{(1-\alpha)}{(1+n)} + \frac{\alpha}{1+r_s}}_{\underbrace{(1+n)}} \begin{bmatrix} \frac{11}{\sum_{i=1}^{\Sigma} r_s P_{kis} r_{cis} dk_{cis} + \sum_{i=1}^{\Sigma} P_{kis} r_{nis} dk_{nis}}_{\underbrace{i=1}^{\Sigma} r_s P_{kis} r_{nis} dk_{nis}} \end{bmatrix}$$

According to equations (47) and (48) the change in utility for the young generation born at time s is  $(1-\alpha)$  times the change in excess burden at time s

plus  $\alpha$  times the present value of the change in excess burden at time s+1, when the young of generation s are old.

We next define the right hand side of (48), which according to (47), is the change in each member of generation s' utility, as  $b_s$  (s\ge t). Also define  $b_{t-1}$  as the right hand side of (45). Define the present value (per initial old person) of the sum (over each generation) of these normalized (by the marginal utility of income) changes in utility as  $dB_t$ , then:

(49) 
$$dB_{t} = \sum_{s=t}^{\infty} b_{s} \frac{(1+n)^{s-t}}{(1+r)^{s-t}} + b_{t-1}$$

To find the value of  $B_t^*$ , the excess burden from shifting from one set of absolute taxes,  $r^*P_1^*\tau_{c1}^*$ 's and  $r^*P_1^*\tau_{n1}^*$ 's, to a new set of absolute taxes,  $\widetilde{rP_1\tau_{c1}}$ 's and  $\widetilde{rP_1\tau_{n1}}$ 's, we define the dummy variable z which runs from 0 through 1. The variable z determines the extent to which the set of tax rates is switched from the \* (initial steady state) set to the new set. For example, we express the rate of tax on corporate industry 1 at time t as  $z\tau_{c1t}^* + (1-z)\tau_{c1t}^*$ . The formula for  $B_t^*$  is given by integrating  $dB_t$  with respect to z over the limits z=0 to z=1. Using the definition of  $b_s$  and  $b_{t-1}$ , the formula for  $B_t^*$  is:

$$(50) \quad B_{t}^{\star} = \sum_{s=t}^{\infty} \int_{0}^{1} \prod_{j=t}^{s} (1+r_{j})^{-1} \begin{bmatrix} 11 \\ \Sigma \\ \underline{i}=1 \end{bmatrix} [z(r^{\star}P_{ki}^{\star}r_{cis}^{\star}) + (1-z)\widetilde{r}\widetilde{P}_{ki}^{\star}\widetilde{r}_{cis}] \frac{\delta K_{cis}}{\delta z} dz$$

$$+ \sum_{s=t}^{\infty} \int_{j=t}^{1} \prod_{j=t}^{s} (1+r_{j})^{-1} \begin{bmatrix} 11 \\ \Sigma \\ \underline{i}=1 \end{bmatrix} [z(r^{\star}P_{ki}^{\star}r_{nis}^{\star}) + (1-z)\widetilde{r}\widetilde{P}_{ki}^{\star}\widetilde{r}_{nis}] \frac{\delta K_{nis}}{\delta z} dz$$

Formula (50) is the intertemporal analogue to Harberger's (1966) excess burden formula. Evaluating the derivatives of the capital stock in (50) at a particular value of z leads to a triangle approximation formula. If we

replace  $\delta K_{\text{cis}}/\delta z$ ,  $\delta K_{\text{nis}}/\delta z$ , and  $r_{\text{s}}$  with their new steady state values, i.e., their values when s is very large, we arrive at the steady state approximation for excess burden proposed by Auerbach (1988, equation 20).

By comparing (48) and (50) one can see that in the new steady state the ratio of B\*, to the present value of the economy's consumption equals the ratio of the change in utility of an individual in the new steady state, which is given by the integral over z of (46) evaluated at the right hand side of (48), divided by the present value of the individual's lifetime consumption. In the steady state the ratio of B\*, to the present value of the economy's consumption equals the ratio of the sum of the integrands in (50), the annual excess burden, divided by annual consumption. But the present value of an individual's lifetime consumption in the new steady state equals the economy's annual consumption per old person divided by the first term in brackets on the right hand side of (48). Hence, except for the fact that the transition may not be immediate, this method of compensation implies that the change in steady state utility divided by the present value of steady state consumption is identical to the present value of the aggregate economy's efficiency gain from tax reform divided by the present value of the aggregate economy's consumption.

The exact formula for excess burden given in equation (50) assumes that the compensation terms  $m_s$  are adjusted as one discretely alters tax rates, i.e., the value of  $dm_s$  depends on the value of z. While (44) is the correct formula for the derivative of  $m_s$ , in order to simulate the model we need a formula for the level of  $m_s$  (the integral of  $dm_s$ ). While we cannot derive this formula analytically, we specify the following approximation formula for  $m_s$ :

$$(51) \qquad m_{s} \stackrel{\sim}{=} \frac{11}{\Sigma} (\widetilde{P}_{i} - P_{i}^{*}) \overline{c}_{12} \stackrel{11}{-\Sigma} (\widetilde{P}_{ki} - P_{ki}^{*}) \overline{k}_{i} \stackrel{11}{-\Sigma} (\widetilde{rP}_{ki} \widetilde{r}_{ci} - r^{*} P_{ki}^{*} r_{ci}^{*}) \overline{k}_{ci}$$

$$= \frac{11}{-\Sigma} (\widetilde{rP}_{ki} \widetilde{r}_{ni} - r^{*} P_{ki}^{*} r_{ni}^{*}) \overline{k}_{ni} \stackrel{11}{-\Sigma} r \overline{P}_{ki} r_{ci}^{*} (\widetilde{k}_{ci} - k_{ci}^{*}) - \frac{11}{\Sigma} r \overline{P}_{ki} r_{ni}^{*} (\widetilde{k}_{ni} - k_{ni}^{*})$$

$$= \frac{11}{-\Sigma} (\widetilde{rP}_{ki} - r_{k}^{*} P_{ki}^{*}) \overline{k}_{i} - \alpha \begin{bmatrix} 11 \\ \Sigma \overline{P}_{i} [\widetilde{x}_{i} - x_{i}^{*}] - \overline{r} & \Sigma \overline{P}_{ki} [\widetilde{k}_{i} - k_{i}^{*}] \end{bmatrix}$$

Since the inital value of  $\mathtt{m_s}$  is zero, the value of  $\mathtt{m_s}$  simply equals the change in  $m_s$ . Equation (51) is a first order approximation (term by term) to (44), which is the formula for very small changes in  $m_{g}$ . In this approximation we evaluate quantities and prices at the average of their initial and final steady state values. In (51) the superscript - denotes average value. The error introduced by using an approximation formula for m, appears to be quite small; while the results we present below are based on (51) which evaluates quantities and prices at their average (as defined above) values, we have also tried modifications of (51) that evaluate both quantities and prices at either their initial or final steady state values. The results are very similar. For example, in all our simulations of the 1986 Tax Reform the difference in the efficiency calculations is less than 1 percent. The difference is also less than 1 percent in all our simulations of the total removal of capital income taxes if we assume a zero intertemporal substitution elasticity. If we assume an intertemporal substitution elasticity equal to .25 the difference is less than 10 percent.

In combination equations (43), (47), and (51) imply:

$$\frac{11}{\overset{\Sigma\overline{P}}{i-1}}(\tilde{c}_{i1}-c_{i1}^{\star})+\frac{1}{1+\overset{\Sigma}{r}}\overset{\Sigma}{\underset{i-1}{r}}\tilde{P}_{i}(\tilde{c}_{i2}-c_{i2}^{\star})=\left[\frac{(1-\alpha)}{(1+n)}+\frac{\alpha}{1+\overset{\Sigma}{r}}\right]\begin{bmatrix}11\\\overset{\Sigma\overline{P}}{\underset{i-1}{r}}[\tilde{x}_{i}-x_{i}^{\star}]&-\overset{11}{\overset{\Sigma}{r}}\overset{\Sigma}{\underset{i-1}{r}}\tilde{k}_{i}[\tilde{k}_{i}-k_{i}^{\star}]\\i=1\end{bmatrix}$$

The left hand side of (52) is a first order approximation to the left hand side of (48), while the right hand side of (52) is a first order approximation to the right hand side of (48). Recall that the left hand side of (48) equals the differential of lifetime utility (see (46)); hence, our first order approximation to m<sub>s</sub> implies that the change in utility of each generation equals, to a first order, the first order approximation to the flow efficiency gain multiplied by the generation's share of consumption.

Our consumption-proportional method of generational compensation is also attractive because it appears to mitigate differences in utility changes over the transition. Indeed, in the case of only a single capital good and a zero intertemporal elasticity of substitution, one can show that, with our compensation scheme, the transition is immediate, i.e., the economy moves immediately to its new steady state. In the one capital good case all generations, including initial generations, experience an identical utility change measured as a percentage of the present value of their remaining lifetime consumption. In the case of more than one capital and a zero intertemporal elasticity of substitution, the transition may not be immediate, but it is likely to be quite quick. The reason is that our compensation scheme, when plugged into the lifetime present value budget constraint (equation 23), implies the following:

(53) 
$$\sum_{i=1}^{11} \overline{P}_{ki} (\widetilde{K}_i - K_i^*) = 0$$

Equation (53) is akin to a fixed capital stock constraint. It says that the change in capital valued at average prices is zero. Clearly, with one capital good, this translates into a fixed capital stock.

While our measure of the efficiency gain from tax reform is based on the steady state change in utility it may overstate or, which seems quite unlikely in our simulations, understate the measure that would arise if we knew the economy's precise transition path. But, as just indicated, if we assume a zero or a small intertemporal elasticity of substitution, our compensation scheme imposes, essentially, a fixed capital stock constraint. Hence, the differences between transitional and new steady state utility levels are likely to be quite small at least in this case. In this regard it is worth pointing out that most of the empirical estimates of the intertemporal elasticity of substitution (e.g., Hall, 1988) indicate that this elasticity is quite small.

### VI. Calibrating and Solving the Model

To solve the model we need to specify functional forms for production, utility, and the distribution of entrepreneurial efficiency coefficients. We assume a CES production function, hence corporate output per-manager in sector i (1-1,10) is given by:

(54) 
$$q_{ci} = (H_i/\mu_i) \begin{bmatrix} (1-a_i - b_i)D_i & (1-1/\sigma_i) & (1-1/\sigma_i) \\ (1-a_i - b_i)D_i & + a_i + a_i \end{bmatrix} (1-1/\sigma_i) + b_i k_{ci} \end{bmatrix}^{\sigma_i/(\sigma_i-1)}$$

In (54)  $\sigma_1$  is the factor substitution elasticity, and  $H_1$ ,  $H_1$ , and  $H_2$  are coefficients. The term  $H_1$  is a coefficient relating the  $H_2$  (value added) function to the QCi function in equation (1), that is  $H_2$  =  $H_2$ . The production function for the noncorporate sector is identical except that  $H_2$  is

replaced by  $\mathbf{A}_1$ , which varies for different entrepreneurs. In the case of owner-occupied housing,  $\mathbf{b}_{11}$  equals 1 and  $\mathbf{a}_{11}$  equals zero.

The joint density function,  $z(A_1, A_2, \ldots A_8)$ , is written as the product of eight independent exponential functions  $\Gamma_1 \exp(-\Gamma_1 A_1)$  times  $\overline{L}$ , the labor force, with  $\overline{A}_1$  set to infinity. The values of the  $\Gamma_1$ 's are determined in the calibration of the model. We also consider the case where the supply of entrepreneurs to each sector is fixed at their initial levels. In this case, the increase in noncorporate production arising from an increase in the corporate tax wedge is due solely to the expansion of output by existing noncorporate firms. In the fixed entrepreneur case there is no need to calculate the values of  $\Gamma_1$  or  $A_1$  since the percentage changes in capital and labor used by entrepreneurs in sector i are identical for all entrepreneurs in sector i and depend solely on changes in output and factor prices.

# A. Calibration of the Model

In the initial observed equilibrium all inputs are measured in units such that all eleven prices, W, r, and all Di's equal unity. The rental rates as expressed in equations (9), (10), and (22) are larger than unity owing to depreciation and taxes; these rental rates in the initial equilibrium are denoted  $R_{ci}^*$  and  $R_{ni}^*$  to distinguish them from the new equilibrium values. With the exception of the input-output coefficients all data used to calibrate the model are contained in Appendix Table A-1. These data include, for each industry, the share of output which is corporate  $(Q_{ci}/Q_i)$ , the distribution of value added across industries,  $s_i$ , the aggregate shares of capital income by industry,  $\beta_i$ , the share of labor income by industry  $\alpha_i$ , corporate and noncorporate effective marginal tax rates by industry,  $r_{ci}$  and  $r_{ni}$ , average and asset-specific depreciation rates by industry, and the composition of

capital by industry. Given the values of depreciation rates, tax rates, and the normalized unitary values of r and all output prices equal, we use equations (9), (10), and (22) to calculate the observed rental rates,  $R_{ci}^*$  and  $R_{ni}^*$ .

Because of tax differentials capital's share of income by industry will (except in the Cobb-Douglas case) differ between the corporate and noncorporate sectors. The labor share of income is the same. These shares are related to the underlying coefficients in the production functions by:

(55) 
$$\alpha_{i} = a_{i}^{\sigma_{i}} H_{i}^{\sigma_{i}-1}$$

(56) 
$$\beta_{ci} = R_{ci}^{\star}^{1-\sigma_i} b_i^{\sigma_i} H_i^{\sigma_i^{-1}}$$

(57) 
$$\beta_{ni} = (R_{ci}^{*}/R_{ni}^{*})^{\sigma}i^{-1}\beta_{ci}$$

where  $\beta_{\text{ci}}$  refers to the corporate share of capital income, and  $\beta_{\text{ni}}$  refers to the noncorporate share of capital income. It is not necessary in calibrating or solving the model to calculate the values of  $\mathbf{a_i}$ ,  $\mathbf{b_i}$ , or  $\mathbf{H_i}$ , since all equilibrium equations can be written in terms of initial income shares.

To proceed further with the calibration of the model requires the functional forms of the relationships between inputs and outputs. The corporate production functions and the three first order conditions result in four equations for each sector. Similarly, the noncorporate production function and its two first order conditions result in three additional equations, and the requirement that profit of the marginal entrepreneur equal the wage rate produces another equation. These eight equations are the first eight equations in Table 2 (equations (58) - (65)), listing functional relationships and providing the specific forms of these variables for use in the equations of Table 1. These forms can apply to both calibration and

solution of the model as W,  $R_{ci}$ , and  $R_{ni}$  refer to the new equilibrium. For purposes of calibrating the model these values are set to 1,  $R_{ci}$ \* and  $R_{ni}$ \* respectively. Owner-occupied housing is treated in the same fashion as corporate production, but only has one first order condition, so that only equation (58) and equation (62) appear for owner occupied housing.

The first step in calibrating the model is to use the observed distribution of aggregate value added, along with the relationships in (56) through (60) to obtain the implied distribution of the aggregate capital stock and the values of the corporate and noncorporate capital income shares. Note that the ratio of  $k_{ci}$  to  $q_{ci}$  is equal to the ratio of aggregate  $K_{ci}$  to  $q_{ci}$  and the ratio of  $k_{ni}(A)$  to  $q_{ni}(A)$  is equal to the ratio of aggregate  $K_{ni}$  to  $q_{ni}$ . Dividing (62) by (63), and rearranging, produces an expression for the share of corporate capital in each industry as a function of output shares.

(67) 
$$K_{ci}/K_{i} = \frac{(Q_{ci}/Q_{i})}{(1-Q_{ci})^{*}(R_{ci}/R_{ni})^{*}} + (Q_{ci}/Q_{i})$$

Since the sum of total capital income,  $\beta_1 Q_1$ , equals the sum of capital income in both sectors, we can also use (61) to obtain the value of  $\beta_{ci}$ .

(68) 
$$\beta_{ci} = \beta_{i} \frac{1}{(Q_{ci}/Q_{i}) + (R_{ci}/R_{ni})} \frac{(\sigma_{i}^{-1})}{(1 - Q_{ci}/Q_{i})}$$

We can similarly use these relationships to obtain the shares of total capital in the eleven industries by using the distribution of value added (value added is  $\mu_4Q_4$ ):

(69) 
$$K_{i}/\bar{K} = \frac{s_{i}\beta_{i}/(K_{ci}R_{ci}^{*}/K_{i}+K_{ni}R_{ni}^{*}/K_{i})}{11} \frac{\sum_{i=1}^{S} s_{i}\beta_{i}/(K_{ci}R_{ci}^{*}/K_{i}+K_{ni}R_{ni}^{*}/K_{i})}{\sum_{i=1}^{S} s_{i}\beta_{i}/(K_{ci}R_{ci}^{*}/K_{i}+K_{ni}R_{ni}^{*}/K_{i})}$$

This process results in all 19 allocations of capital (10 Kci's, 8 Kni's and Kll) as a share of an as yet unknown total capital stock plus all of the values of the initial capital income shares.

Equations (58) and (59), which relate value added to capital, can be used to calibrate the initial total level of the capital stock, given that the sum of value added is equal to GNP. Thus, the sums of value added are rewritten as the sums of aggregate capital stocks multiplied by functional relationships. Dividing by total capital,  $\overline{K}$ , results in an equation for the total capital stock as a function of prices, capital shares, and GNP. Once total capital is measured, the total levels of the 19 capital stocks can be determined. Next we can measure  $M_{ci}$  (equal to  $K_{ci}/k_{ci}$ ),  $L_{ci}$  (equal to  $M_{ci}l_{ci}$ ), and  $L_{ni}$  (since  $L_{ni}/K_{ni}$  equals  $l_{ni}(A)/k_{ni}(A)$ , via equations (60),(61) and (65).

The next step in calibrating the model is to determine the total labor force  $\overline{L}$  and the eight values of  $\Gamma_1$ . Total  $\overline{L}$  is the sum of the already measured values of  $M_{ci}$ ,  $L_{ci}$  and  $L_{ni}$  and the entrepreneurs. The integral in equation (28) contains two parts, the first measuring the entrepreneurs and the second noncorporate labor. Consider the first part, and substitute the exponential density function described above in the integral measuring entrepreneurs for sector 1. The solution to this integral produces a series of 128 terms all of the form  $e^{-x}/x$ . The first, positive term in the series is  $x = \Gamma_1 A_1$ , the next seven terms are all of the form  $x = \Gamma_1 A_1 + \Gamma_2 A_2$ ,  $\Gamma_1 A_1 + \Gamma_3 A_3$ , ... and enter with a negative sign. The next 21 terms, which enter with a positive sign are the sum of  $\Gamma_1 A_1$  and all unique combinations of two of the remaining coefficients, i.e,  $\Gamma_1 A_1 + \Gamma_2 A_2 + \Gamma_3 A_3$ ,  $\Gamma_1 A_1 + \Gamma_2 A_2 + \Gamma_4 A_4$ , .... There are then 35 terms (with negative sign) which include unique combinations

of three of the remaining sectors, 35 terms with positive sign which include unique combinations of 4, 21 terms with negative sign which include unique combinations of 5, 7 with positive signs which include unique combinations of 6, and the final term with a negative sign which is the sum of all terms,i.e.,  $\Gamma_1 A_1 + \Gamma_2 A_2 + \Gamma_3 A_3 + \Gamma_4 A_4 + \Gamma_5 A_5 + \Gamma_6 A_6 + \Gamma_7 A_7 + \Gamma_8 A_8$ . These 128 terms are added, and the total is, in turn, multiplied by  $LA_1$ . A similar form occurs for the other seven sectors. Since all of these are multiplied by  $\overline{L}$  and since total  $\overline{L}$  is the sum of these entrepreneurs and other labor, we can obtain a solution for  $\overline{L}$  of the form:

(70) 
$$\overline{L} = \begin{bmatrix} 11 & 11 & 11 & 11 & 8 & -\Gamma_{1}A_{1} \\ \sum_{i=1}^{L} C_{i} + \sum_{i=1}^{L} M_{C_{i}} + \sum_{i=1}^{L} L_{n_{i}} \end{bmatrix} / \frac{8}{\pi} (1-e^{-\Gamma_{1}A_{1}})$$

When summing, for each sector, capital in the non-corporate sector there is an  $A_i$  term multiplying the density function. (The remaining terms for  $k_{ni}(A_i)$  relating to factor shares and prices can be taken outside the integrals as constants.) The solution to this integral also involves a series of 128 terms except that the functional form is now  $e^{-x}(1/x + 1/x^2)$ . The sum of this series of terms is all multiplied for each sector by  $IA_i\Gamma_i$ , but  $\overline{L}$ can now be written using equation (70). Since total  $K_{ni}$  (which is known) is equal to the integral of  $k_{ni}(A_i)$  we now have eight equations in the eight unknown values of  $\Gamma_i$ , which can be solved simultaneously. Once these values are known we can calculate the value of  $\overline{L}$  from (70) and the number of entrepreneurs in each sector.

Another version of the model is calibrated under the assumption that entrepreneurs in each sector are in fixed supply. In this case the sum of  $\mathbb{M}_{ci}$ ,  $\mathbb{L}_{ci}$ , and  $\mathbb{L}_{ni}$  is held constant (since that will lead to a fixed labor supply with fixed entrepreneurs) and there are no distributional parameters in

the model. Rather the value of each integral for capital is measured as a fixed quantity (only the functional terms containing prices which can be brought outside the integral vary).

At this point all the levels of labor, capital and output are determined, and we can use equations (32) through (34) to determine the amounts of consumption for each of the eleven commodities. The eleven first order conditions from the utility function can be used to form the ten ratios of consumption which along with holding of the sum of consumption expenditure constant lead to the functional forms of the ratios of consumption of each good to total consumption in equation (66). These eleven equations can be used to solve for the eleven g's.

The final parameter value needed is the rate of time preference. By combining the relationships between observed patterns of consumption with the growth rate and interest rate, we have, in our 55 year continuous time model, the steady state equation:

$$\frac{\frac{11}{121}P_{1}C_{1}}{\sum_{i=1}^{21}\mu_{i}P_{i}Q_{i}-R_{ci}K_{ci}-R_{ni}K_{ni}} = \frac{\left(1-e^{-(n+\rho-r\eta)^{55}}\right)\left(1-e^{-r^{g40}}\right)\left(\rho+r^{g}-r\eta\right)n}{\left(1-e^{-(\rho+r^{g}-r\eta)^{55}}\right)\left(1-e^{-n^{40}}\right)\left(n+\rho-r\eta\right)r^{g}}$$

In a model without rents, the denominator on the left hand side of the equation would simply be WL; since there are rents in this model, we must calculate labor income as a residual of GNP minus capital income. This equation is used to calibrate the time preference factor  $\rho$ . Note that while we are free to set r to any value (including unity) by defining capital in units larger than a dollar, (and must also conform measures of depreciation flows and net investment flows accordingly) in the present value formulation r and n must represent pure rates. This is a simple procedure. If r measured

as a rate is .05 and our units of capital are measured as twenty dollar units, the pre-tax, net of depreciation, rental must be divided by twenty to produce r as a rate per dollar.

### B. Solving the Model

To solve the model, when a new set of tax rates is introduced we guess values of r and W. Using the eleven forms of equation (35) as modified by equation (62), we then solve for the eleven prices which are consistent with these values. Because the model includes capital asset prices, these equations are non-linear and must be solved numerically. The new set of prices along with the new tax rates are used to calculate the values of Rci and Rni, and, with  $R_{ci}$  and  $R_{ni}$ , the values of  $A_i$ . Next we solve for the amounts of noncorporate capital, the levels of noncorporate output, and the number of noncorporate workers and entrepreneurs, all of which depend solely on the calibrated parameters and the levels of r, W, prices, and tax rates. Given the prices and the numeraire convention that fixes total expenditures on consumption, we can also determine the consumption of each good.

Since corporate capital,  $K_{ci}$ , can be written in terms of corporate output  $Q_{ci}$ , via equation (58), the eleven supply and demand equations in (32)- (34) can be used to solve for the eleven values of  $Q_{ci}$  (treating owner-occupied housing in the same manner as corporate output), and in turn for the values of  $K_{ci}$ ,  $M_{ci}$ , and  $L_{ci}$ . All of the allocations of capital, labor and output have now been made. If we have guessed the equilibrium values for r and W, the resulting allocations will satisfy the labor constraint in (28) and the budget constraint that arises after substituting in the compensation scheme. This constraint is:

(72)

$$\frac{(n+\rho-\bar{r}_{\eta})\left(1-e^{(\rho+\bar{r}_{1}-\bar{r}_{\eta})55}\right)}{\left(1-e^{-(n+\rho-\bar{r}_{\eta})55}\right)\left(\rho+\bar{r}_{1}-\bar{r}_{\eta}\right)} \underbrace{\Sigma\bar{P}_{i}\bar{C}_{i}}_{=\frac{(n+\rho-r_{\eta})\left(1-e^{-(\rho+\bar{r}_{1}-r_{\eta})55}\right)}{\left(1-e^{-(n+\rho-r_{\eta})55}\right)\left(\rho+\bar{r}_{1}-r_{\eta}\right)}} \underbrace{\Sigma\bar{P}_{i}\bar{C}_{i}}_{=\frac{(n+\rho-r_{\eta})55}{(n-r_{\eta})55}} \underbrace{\Sigma\bar{P}_{i}\bar{C}_{i}}_{=\frac{(n+\rho-r_{\eta})55}{(n-$$

If the guessed values of r and W do not satisfy the labor constraint and the budget constraint, we use these two equations to update our guesses of r and W and continue to iterate on the values of r and W until we have reached convergence. Typically about 10 iterations are required for convergence to five decimal places. We have encountered no difficulties in finding an equilibrium.

# VII. The Findings

### A. The Tax Reform Act of 1986

The Tax Reform Act of 1986 had a relatively small effect on the overall marginal effective tax on capital income. At the pre-Tax Reform allocations of capital the overall effective tax rate is estimated at 32 percent prior to the Tax Reform Act and 31 percent after the Tax Reform Act. There were, however, major changes in the allocation of tax burdens across assets and sectors. The statutory corporate tax rate was reduced from 46 percent to 34 percent, and the average marginal personal tax rate on capital income fell from about 30 percent to about 23 percent due to statutory rate reductions. As the price for these rate reductions the investment tax credit, which primarily benefited equipment and public utility structures, was repealed and depreciation was made somewhat less liberal, particularly for structures. Hence, effective marginal tax rates were increased for equipment and lowered for other assets, particularly land and inventories where there was no loss of

marginal tax incentives to offset the reduced statutory rates. With respect to the corporate tax wedge, the investment tax credit's repeal had a small effect, but the reduction in the statutory corporate tax rate had a significant impact. Because of the changes in relative corporate non-corporate effective marginal tax rates corporate tax wedges were increased in some industries and reduced in others, although in most cases they were reduced. The corporate tax wedges for the eight sectors are shown in Table 3.

The second reason that the Tax Reform Act reduced the distortions from the corporate tax wedge is that industries with larger shares of noncorporate production typically experienced the largest reductions in the corporate tax wedge. Rental housing uses only land and structures; agriculture is also a very intensive user of land. These two industries account for 64 percent of noncorporate capital, and their tax wedges were reduced, respectively, from .59 to .42 and from .69 to .49. The Trade industry also experienced a significant reduction in its tax wedge, from .69 to .54, reflecting the importance of inventories in its capital stock. These three industries, rental housing, agriculture, and trade, accounted for 77 percent of noncorporate capital. Only services, which has virtually no inventories, experienced an increase in its corporate tax wedge, from .41 to .49. Of the remaining sectors the corporate tax wedge change was either quite small or increased, but there is very little noncorporate capital in these sectors.

Table 4 presents our efficiency gain findings. We measure the efficiency gain as the compensating variation in the initial steady state needed to obtain the final steady state level of utility divided by the initial steady state present value of lifetime consumption expenditure. We report both efficiency gains from the 1986 Tax Reform Act and the efficiency gains from the complete elimination of capital income taxation. Since our main focus is

on the corporate non-corporate tax distortion, we consider a zero intertemporal substitution elasticity as well as a value of .25 for this elasticity. Since there is no intertemporal distortion if the intertemporal elasticity is zero, the reported efficiency gain in this case will solely reflect corporate non-corporate and interindustry tax wedges. We view the results based on the assumption of fixed entrepreneurs as the most reliable since the results based on assuming variable entrepreneurs require invoking our distribution assumptions on entrepreneurial abilities which may or may not be fully appropriate. The variable enterpreneur case also leads to much larger shifts in corporate shares of sector output than seem reasonable.

The first five rows of Table 4 consider the efficiency gain from tax reform in the case of fixed entrepreneurs, but under alternative parameter assumptions. In each case the efficiency gain is about four fifths of one percent of the present value of consumption. This is a reasonably large figure when compared with the efficiency gains from alternative tax reforms discussed in the public finance literature. Given the 1988 value of aggregate U.S. consumption, this efficiency gain corresponds to roughly \$31 billion per year, a quite significant amount. As one would expect, the assumption of variable entrepreneurs raises the estimated efficiency gain considerably; indeed, it more than doubles it.

Since the 1986 Tax Reform Act did not greatly alter the overall effective marginal tax on capital income, it is not surprising that the efficiency gains from the reform are very similar in the cases of zero and .25 intertemporal elasticities of substitution. There is a larger difference, however, if capital income taxes are completely removed. Rows 7 and 8 indicate that the efficiency gain from eliminating all capital income taxes is 1.91 percent of consumption in the case of a zero intertemporal elasticity and 2.15 in the

case of a .25 intertemporal elasticity. These figures suggest that most of the excess burden from capital income taxes in the U.S. is due not to the intertemporal distortion, but rather to the distortion in the allocation of the use of capital at a point in time. This finding, while based on a quite different model than Chamley's (1981), accords with Chamley's assessment of the relative size of intertemporal and static distortions associated with capital income taxation.

The large efficiency gains in our model from eliminating the corporate tax wedge reflect significant increases in the corporate share of sector output in those sectors that originally are significantly noncorporate. In particular our simulations predict that the rental housing industry becomes substantially more corporate. While the predicted increase in corporate capital in all the other sectors seems reasonable by historic standards, the shift predicted for rental housing is quite large. Gravelle (1989b) modifies the model to disaggregate rental housing into a multi-family rental housing sector with corporate as well as non-corporate production and a single family rental housing sector which is solely non-corporate. This modification limits the reallocation of rental housing toward corporate housing and reduces the estimated efficiency gain from the 1986 Tax Reform by about one half, to \$15 billion dollars per year. This efficiency gain, amounting to .43 percent of 1988 U.S. consumption is, nonetheless, quite susbstantial. It is over five times the corresponding efficiency gain predicted by the Harberger model.

#### VIII. Conclusion

This paper has shown that how one models the corporate non-corporate tax distortion is crucial for assessing the efficiency effects of the 1986 Tax Reform Act and presumably, other past and future tax reforms as well. While

our basic finding of a \$31 billion annual efficiency gain may be viewed as somewhat high, given its dependence on a significant reincorporation of the rental housing industry, even the \$15 billion figure that results from a more conservative treatment of rental housing is substantial. In addition to demonstrating that the corporate non-corporate distortion in the U.S. may be much more serious than previously believed, this paper suggests that the intratemporal distortion in the allocation of capital is more serious that the intertemporal distortion.

The findings have important implications for the design of tax structures. For example, if the major source of capital income tax distortion involves the corporate non-corporate tax wedge, integration of the corporate and individual income taxes is a policy option deserving of much more attention. In comparison with policies that reduce the corporate tax wedge, policies, such as changes in the investment tax credit, which primarily affect the intertemporal distortion, may be less effective in reducing the excess burden from capital income taxation.

## Table 1

## The Steady State Model

(29) 
$$K_{i} = M_{ci}k_{ci} + \int_{\underline{A}_{i}}^{\overline{A}_{i}} \prod_{j \neq i} \int_{0}^{A_{i}\underline{A}_{-j}/\underline{A}_{-i}} k_{ni}(A_{i}) z(A_{1}, A_{2}, ..., A_{8}) dA_{1}dA_{2}...dA_{8}$$
  $i=1,10$ 

$$\text{(30)} \quad Q_{i} = \text{M}_{ci} q_{ni} + \sum_{\underline{A}_{i}}^{\overline{A}_{i}} \prod_{j=1}^{\overline{A}_{i}} \int_{0}^{A_{i} \underline{A}_{j} / \underline{A}_{i}} q_{ni} (A_{i}) \ z(A_{1}, A_{2}, \dots, A_{8}) \ dA_{1} dA_{2} \dots dA_{8} \qquad i=1,10$$

(31) 
$$c_{i} = \sum_{j=1}^{55} \frac{c_{j,i}}{(1+n)^{j-1}}$$
  $i=1,11$ 

(32) 
$$Q_{i} = C_{i} + \sum_{j=1}^{11} v_{ij}Q_{j} + n\omega_{i}K_{i}$$
  $i \neq 5, 9$ 

(33) 
$$Q_{5} = C_{5} + \sum_{j=1}^{11} v_{5j}Q_{j} + n\omega_{5}K_{5} + \sum_{j=1}^{11} (n+\delta_{5j})\psi_{j}K_{j}$$

(34) 
$$Q_{9} - C_{9} + \sum_{j=1}^{11} v_{9j} Q_{j} + n \omega_{9} K_{9} + \sum_{j=1}^{11} (n + \delta_{9j}) \xi_{j} K_{j}$$

# Table 1 Continued

(35) 
$$P_{i} = \sum_{j=1}^{10} v_{ji} P_{j} + R_{ci} k_{ci} / q_{ci} + W(1+1_{ci}) / q_{ci}$$
 i=1,10

$$(36) \qquad \underline{A_i} x_i (W, R_{ni}, P_i) = W \qquad i=1,8$$

(37) 
$$\frac{c_{j,k}}{c_{j,m}} = \begin{bmatrix} \frac{P_k}{P_m} \end{bmatrix}^{-\varphi}$$
  $j=1,2, k=1,10, m(\neq k)=1,10$ 

(38) 
$$\frac{c_{j,k}}{c_{j,11}} = \left[\frac{P_k}{R_{11}}\right]^{-\varphi}$$
 j=1,2 k=1,1

(39) 
$$c_{j+1,i} - c_{j,i} (1+r)^{\eta}/(1+\rho)$$
  $j+1,2$   $i=1,1$ 

(41) 
$$\sum_{i=1}^{11} P_i C_i = \overline{I}$$

Table 2: Equations Used to Calibrate the Model

(58) 
$$\left[ \left( k_{ci} / q_{ci} \right) - \left( \mu_{i} \beta_{ci} / R_{ci} \right) \left[ \left( 1 - \beta_{ci} \right) \left( W R_{ci}^{\star} / R_{ci} \right)^{(1 - \sigma_{i})} + \beta_{ci} \right]^{\sigma_{i} / (1 - \sigma_{i})} \right]$$

(59) 
$$k_{ni}(A_i)/q_{ni}(A_i) - (k_{ci}/q_{ci})(R_{ci}/R_{ni})^{\sigma}$$

(60) 
$$k_{ci} - (\beta_{ci}/(1-\alpha_{i}-\beta_{ci})^{\sigma_{i}} (W/R_{ci})^{\sigma_{i}} R_{ci}^{\star(\sigma_{i}-1)}$$

$$(61) 1_{ci} - \alpha_{ci} / \left[ 1 - \alpha_i - \beta_{ci} \right]$$

(62) 
$$R_{ci}^{k}_{ci}/q_{ci} + W(1+1_{ci})/q_{ci} - \mu_{i} \left[ (1-\beta_{ci})W^{(1-\sigma_{i})} + \beta_{ci}(R_{ci}/R_{ci}^{*})^{*} \right]^{(1/(1-\sigma_{i})}$$

$$(63) \ \underline{A}_{i} = \left[1 - \left[\beta_{ci}/(1 - \alpha_{i} - \beta_{ci})\right] \left[\left(R_{ci}/R_{ci}^{\star}W\right)^{(1 - \sigma_{i})} - \left(R_{ni}/R_{ci}^{\star}W\right)^{(1 - \sigma_{i})}\right]\right]^{-1/(1 - \sigma_{i})}$$

$$(64) \ k_{ni}(A_{i}) = A_{i} \left[ \left( \beta_{ci} / (R_{ci}^{\star} (1 - \alpha_{i} - \beta_{ci})) \right) x \right]$$

$$\left( (R_{ci}^{\star} W / R_{ci})^{(1 - \sigma_{i})} + \left( \beta_{ci} / (1 - \alpha_{\overline{i}} \beta_{ci}) \right) (R_{ci} / R_{ni})^{(1 - \sigma_{i})} \right) \right]^{\sigma_{i} / (1 - \sigma_{i})}$$

$$1_{ni}(A_i) - k_{ni}(A_i)(\alpha_i/\beta_{ci})R_{ci}^{*(1-\sigma_i)}(R_{ni}/W)^{\sigma_i}$$

(66) 
$$C_{i} - g_{i}^{\phi} P_{i}^{\phi} / \left( \sum_{i=1}^{11} g_{i}^{\phi} P_{i}^{(1-\phi)} \right)$$

Table 3
Corporate Tax Wedges by Sector

Sector	<u>Be f</u>	ore Tax	Reform	After Tax Reform			
# Name	$r_{ci}$	$r_{ ext{ni}}$	rci-rni	<sup>τ</sup> ci_	"ni	<sup>τ</sup> ci <sup>-τ</sup> ni	
<u> </u>							
1 Rental Housing	.887	. 299	. 588	.667	. 250	.416	
2 Agriculture	1.083	. 389	. 694	.7 <b>8</b> 6	. 299	. 487	
3 Oil and Gas	. 515	. 124	. 392	.493	. 149	. 343	
4 Mining	.587	. 250	. 337	.724	. 250	.474	
5 Construction	.613	.099	. 514	.786	. 299	.487	
6 Transportation	. 515	. 124	. 392	. 695	. 235	.460	
7 Trade	1.083	. 389	. 694	.887	. 351	.535	
8 Services	. 695	. 282	. 412	. 786	. 299	.487	
9 Manufacturing	. 961			.852			
10 Utilities	. 587			.667			
11 Owner-Occupied Housing	.042			. 042			

Table 4 Efficiency Gains from the 1986 Tax Reform Act and the Complete Elimination of Capital Income Taxation

<u>σ2</u>	<u>g</u>	<u>_n_</u>	<u>F/V</u>	T/N	Efficiency Gain <sup>1</sup>
1	1	0	F	T	. 85
1	1	. 25	F	T	. 84
1	. 5	0	F	T	. 82
1.25	1	0	F	T	.85
.75	. 1	0	F	Т	.81
1	1	0	v	T	1.99
1	1	0	F	N	1.96
1	1	. 25	F	N	2.23
1	. 5	0	F	N	1.77
1.25	1	0	F	N	2.07
.75	1	0	F	N	1.81
1	1	0	v	N	2.97
	1 1 1.25 .75 1 1 1 1.25	1 1 1 1 1 1 .5 1 .75 1 1 1 1 1 1 1 1 1 .5 1 .75 1 .75 1	1     1     0       1     1     .25       1     .5     0       1.25     1     0       .75     1     0       1     1     0       1     1     .25       1     .5     0       1.25     1     0       .75     1     0	1       1       0       F         1       1       .25       F         1       .5       0       F         1.25       1       0       F         .75       1       0       F         1       1       0       F         1       1       .25       F         1       .5       0       F         1.25       1       0       F         .75       1       0       F	1 1 0 F T 1 1 .25 F T 1 .5 0 F T 1.25 1 0 F T .75 1 0 F T 1 1 0 V T 1 1 0 F N 1 1 .25 F N 1 .5 0 F N 1 .5 0 F N 1 .5 0 F N 1.25 1 0 F N 1.25 1 0 F N

<sup>1.</sup> Efficiency gain is the steady state compensating variation in utility divided by the initial steady state present value of lifetime consumption.

 $<sup>\</sup>sigma_1$  - factor substitution elasticity in rental housing and agriculture.

 $<sup>\</sup>sigma_2^{\rm 1}$  = factor substitution elasticity in all other sectors.  $\varphi$  = intratemporal demand elasticity.

 $<sup>\</sup>nu$  = intertemporal demand elasiticity

F = fixed entrepreneurs.

V = variable entrepreneurs.

T - Tax Reform Act.

N = no tax.

### Notes

- 1. The models of Harberger (1966) and Shoven (1976) do not allow for corporate/non-corporate substitution within an industry, intermediate goods, or the production, pricing, and depreciation of capital. They also rely on average rather than marginal tax rates. The models of Fullerton, Shoven and Whalley (1978, 1983), and Ballard, Fullerton, Shoven, and Whalley (1985) introduce intermediate goods and considerable additional detail to their model that permit the measurement of other distortions. Fullerton and Henderson (1989) also allow a mechanism for substitution between corporate and noncorporate capital in different industries. They incorporate depreciation to the extent of adjusting for depreciation in calculating marginal effective tax rates. Their production functions are, however, production functions for output net of depreciation, and there are no asset prices or asset markets.
- 2. Auerbach distributes efficiency gains to each initial young generation, thereby leaving the initial old generation's welfare unchanged. In contrast, we allocate the efficiency gain all generations, including the initial old generation, in proportion to their initial consumption. Clearly, the steady state change in utility arising from a tax reform will differ as between Auerbach's method and our own. The advantage of distributing the efficiency gain in proportion to consumption is that the steady state change in utility, measured as a fraction of the present value of steady state consumption, will equal the steady state value of the present value of the economy's current plus future efficiency gain divided by the present value of the economy's current plus future consumption; i.e., one can read off the aggregate economy's efficiency gain from tax reform by simply considering what happens to individual steady state welfare. Gravelle (1989a) also discusses the issue of how to allocate efficiency gains.
- 3. After developing our model we became aware of Lucas' (1971) paper that also models entrepreneurs as managers with differing abilities and demonstrates how such a model can explain secular changes in firm size. Chamley (1983) is another example of an early analysis of differing entrepreneurial abilities and the choice of occupation.
- 4. Since aggregate labor supply is inelastic in this model and labor taxes are compensated back in a lump sum, labor taxes have no effect on the equilibrium. We therefore leave them out of the model.
- 5. Recall that personal and corporate capital income taxes are rebated back to the owners of capital at the time the capital income is earned. In addition, we value capital at replacement cost which is equivalent to valuing capital at market value and lump sum rebating any changes in capital value associated with tax changes, such as changes in the investment tax credit, back to the owners of capital. Also note that if each old person receives  $\mathbf{m}_{\mathrm{S}}(1+\mathrm{n})$  at time s, the total amount paid by each of the young at time s is  $\mathbf{m}_{\mathrm{S}}$ .

#### References

Auerbach, Alan J., "The Deadweight Loss from 'Nonneutral' Capital Income Taxation," National Bureau of Economic Research Working Paper No. 2510, February 1988.

Auerbach, Alan and Laurence J. Kotlikoff, <u>Dynamic Fiscal Policy</u>, Cambridge, England: Cambridge University Press, 1987.

Bailey, Martin J. "Capital Gains and Income Taxation." <u>In The Taxation of Income from Capital</u>, ed. by Arnold C. Harberger and Martin J. Bailey, Washington, D.C. The Brookings Institution, 1969.

Ballard, Charles L., Don Fullerton, John B. Shoven, and John Whalley.  $\underline{A}$  General Equilibrium Model for Tax Policy Evaluation, Chicago: University of Chicago Press, 1985.

Chamley, Christophe, "Entrepreneurial Abilities and Liabilities in a Model of Self Selection," <u>Bell Journal of Economics</u>, vol. 14, Spring 1983, 70-80.

Ebrill, Liam P. and David G. Hartman, "On the Incidence and Excess Burden of the Corporation Income Tax", <u>Public Finance</u>, 37, 1982.

Eisner, Robert, "Capital Gains and Income: Real Changes in the Value of Capital in the United States, 1946-77," in <u>The Measurement of Capital</u>, ed. by Dan Usher, Chicago, University of Chicago Press, 1980.

Fullerton, Donald, Robert Gillette, and James Mackie, "Investment Incentives Under the Tax Reform Act of 1986," <u>Compendium of Tax Research 1978</u>, Office of Tax Analysis, 1987, pp.131-171.

Fullerton, Don, and Yolanda K. Henderson. "A Disaggregate General Equilibrium Model of the Tax Distortions Among Assets, Sectors, and Industries," <a href="International Economic Review">International Economic Review</a> 30 (May 1989): 391-413.

Fullerton, Don, Yolanda K. Henderson, and James Mackie, "Investment Allocation and Growth Under the Tax Reform Act of 1986," in <u>Compendium of Tax Research 1987</u>, <u>Office of Tax Analysis</u>, Washington, D.C., U.S. Government Printing Office, 1987.

Fullerton, Don, John B. Shoven, and John Whalley, "General Equilibrium Analysis of U.S. Tax Policy," in <u>1978 Compendium of Tax Research, Office of Tax Analysis</u>, Washington, D.C., U.S. Government Printing Office, 1978

Fullerton, Don, John B. Shoven, and John Whalley, "Replacing the U.S. Income Tax with a Progressive Consumption Tax," <u>Journal of Public Economics</u> 20, 1983, pp. 3-23.

Gravelle, Jane G. "Capital Income Taxation and Efficiency in the Allocation of Investment," <u>National Tax Journal</u>, vol. 36, September, 1983: 297-306.

Gravelle, Jane G., "Effects of the 1981 Depreciation Revisions on the Taxation of Income from Business Capital," <u>National Tax Journal</u> vol. 35, March, 1982: 1-20.

Gravelle, Jane G. (1989a), "Income, Consumption and Wage Taxation in a Life Cycle Model: Separating Efficiency from Redistribution." mimeo, June 1989.

Gravelle, Jane G. (1989b), "Differential Taxation of Capital Income: Another Look at the Tax Reform Act." forthcoming in <u>The National Tax Journal</u>, December 1989.

Gravelle, Jane G. and Laurence J. Kotlikoff, (1988a). "Does the Harberger Model Greatly Understate the Excess Burden of the Corporate Tax? - Another Model Says Yes," National Bureau of Economic Research working paper 2742, 1988.

Gravelle, Jane G. and Laurence J. Kotlikoff, "The Incidence and Efficiency Costs of Corporate Taxation when Corporate and Noncorporate Firms Produce the Same Good," <u>Journal of Political Economy</u>, vol. 97, August 1989: 749-780.

Hall, Robert E., "Intertemporal Substitution in Consumption," <u>Journal of Political Economy</u>, vol. 96, April 1988: 339-357.

Hall, Robert and Dale W. Jorgenson. "Tax Policy and Investment Behavior." American Economic Review vol. 57, 1967: 391-414.

Harberger, Arnold, "Efficiency Effects of Taxes on Income from Capital," in <u>Effects of the Corporation Income Tax</u>, ed. by Marianne Krzyzniak. Detroit: Wayne State University Press, 1966.

Harberger, Arnold, "The Incidence of the Corporation Income Tax," <u>Journal of Political Economy</u>, 70 (3), June 1962.

Harberger, Arnold C. "Taxation, Resource Allocation, and Welfare," in <u>The Role of Direct and Indirect Taxes in the Federal Reserve System.</u> Princeton: Princeton University Press, 1964.

Harberger, Arnold C. and Neil Bruce, "The Incidence and Efficiency Effects of Taxes on Income from Capital," <u>Journal of Political Economy</u> 84 (December 1976): 1285-1292.

Hendershott, Patric and Sheng Cheng Hu, "Investment in Producer's Durable Equipment." in <u>How Taxes Affect Economic Behavior</u>, ed. by Henry J. Aaron and Joseph A. Pechman, Brookings Institution, 1981.

Hulten, Charles R. and Wykoff, Frank C. "The Measurement of Economic Depreciation," in <u>Depreciation</u>. <u>Inflation and the Taxation of Income from Capital</u>, ed. by Charles R. Hulten, Washington, D.C., The Urban Institute, 1981.

Rosenberg, Leon G., "Taxation of Income from Capital by Industry Group," in The Taxation of Income from Capital, ed. by Arnold C. Harbergwer and Martin J. Bailey, Washington, D.C.: Brookings Institution, 1969.

- Shoven, John B., "The Incidence and Efficiency Effects of Taxes on Income from Capital," <u>Journal of Political Economy</u>, vol. 84, no. 6, December 1976.
- U.S. Department of Commerce, "Input-Output Accounts of the U.S. Economy, 1981," <u>Survey of Current Business</u>, January 1987, pp. 42-57.
- U. S. Department of Treasury. Internal Revenue Service. <u>Statistics of Income</u>, U.S. Government Printing Office, Washington, D.C., various years.

## Appendix: Data Sources

The model is calibrated using data on corporate shares of productions, industry shares of total value added, aggregate shares of income accruing to capital income and to workers, depreciation rates, and the composition of the capital stock in each industry, as well as estimated initial tax rates. Initial prices, wage rates, and net returns to capital are set to 1. Table Al lists the data used to calibrate the model with the exception of the inputoutput coefficients. Data sources differ slightly with respect to their dates, although all come from the early 1980's. This reflects differences in data availability. Shares of corporate output are taken from Gravelle and Kotlikoff (1989). Shares of value added and capital income shares were obtained from the National Income and Product Accounts for 1982. Capital income shares are the sum of depreciation, profits, interest, and property taxes divided by value added. Rosenberg's (1969) data were used to allocate property taxes and proprietorship income to capital and labor. Input-output data are from the latest input-output accounts, 1981 reported in The Survey of Current Business, January 1987. Labor income shares are estimated from data on payments to labor from sole proprietorship tax returns for 1980, the last year in which such data were reported for agriculture.

The derivation of marginal tax rates requires a detailed description. In earlier studies, such as Shoven (1976), the level of tax burden was typically judged by current (the cash flow of) taxes divided by some measure of current income. These average tax rates did not properly account for the advantage of the timing of tax write offs, such as depreciation allowances, and they failed to distinquish taxation of the existing stock of capital from taxation of new investment. As is well known, average tax rates based on the cash flow of taxes have no necessary relationship to marginal effective tax rates on new

investment, and it is the latter that governs the efficiency effects of tax policies.

Our marginal effective tax rates are based on a discounted cash flow formula which compares the internal rate of return with and without taxes. This measure accounts for all of the timing effects associated with certain tax preferences, including accelerated depreciation and deferral of taxes on capital gains until realization. Most current studies, such as Fullerton, Henderson, and Mackie (1987) and Gravelle (1982) employ such marginal tax rates.

In the case of a depreciating asset, the relationship between the pre-tax return and the after-tax return in the corporate sector is determined by the rental price formula of Hall and Jorgenson (1967):

(A1) 
$$r_p = ((r_f + \delta)(1 - t_f PVD(1 - mk) - k))/(1 - t_f) - \delta$$

where  $r_p$  is the pre-tax real return,  $r_f$  is the after-tax discount rate of the firm,  $\delta$  is the economic depreciation rate,  $t_f$  is the statutory tax rate of the firm (equal to the corporate tax rate for corporate production and equal to the individual tax rate for noncorporate production), PVD is the present value of depreciation deductions for tax purposes, k is the investment tax credit rate, and m is the fraction of k that reduces the basis for depreciation purposes. Note that t refers to the tax on gross income as more commonly used; it is related to  $\tau$  by  $t-\tau/(1-\tau)$ . The value of depreciation is discounted at the nominal discount rate,  $r_f + \pi$ , where  $\pi$  is the rate of inflation. This formula applies to investments in equipment and structures which are subject to depreciation.

In the case of an appreciating asset, assuming that returns are not indexed:

(A2) 
$$e^{(r_f^{+\pi})T} - e^{(r_p^{+\pi})T} (1-t_f) + t_f$$

where T is the holding period. This type of calculation applies to inventories at the level of the firm, when FIFO (first-in, first-out) accounting is used. When LIFO (last-in, first-out) accounting is used,  $\pi$  is set to zero in equation (A2). This formula simply expresses the mathematical relationship between pre-tax and after-tax returns when income is taxed on a realization basis. This approach treats inventories as goods under construction whose accrued value is not taxed until the product is sold; this deferral aspect is relatively unimportant in the case of inventories where holding periods are very short; the same type of formula is used, however, in the calculation of effective capital gains tax rates on corporate equity where deferral of tax is important.

Discount rates will differ between the corporate and noncorporate sector. For the noncorporate firm, the discount rate is the overall net after-tax return, r. In the case of the corporate sector, however, there is another layer of tax imposed, so that  $\mathbf{r_f}$  for the corporate firm is different from r. Corporations are allowed to deduct interest at their higher statutory tax rate, including the inflation premium and this interest is subject to individual tax at the personal level. In addition, the equity return to capital is subject to tax as dividends and capital gains. The discount rate of the corporations is equal to a weighted average of debt and equity:

(A3) 
$$r_f = f(i(1-t_f)-\pi) + (1-f)E$$

where f is the fraction financed by debt, i is the nominal interest rate,  $\tau_{\mathrm{f}}$  is the corporate statutory rate, and E is the return required by stockholders prior to personal level taxes. The discount rate of the noncorporate firm is:

(A4) 
$$r = f(i(1-t_f)-\pi) + (1-f)E(1-v)$$

where  $r_{\mathbf{f}}$  is this formula is the individual tax rate and v is the effective tax rate on corporate equity at the personal level. This formula simply says that the overall net after-tax return is a weighted average of the after-tax return to debt and the after-tax return to equity, assuming individuals must earn the same return on their equity investments in the noncorporate sector as in the corporate sector. The value of v is determined by the formula:

(A5) 
$$E(1-v) = sdE(1-t_f) + (1n(e^{(E(1-sd)+\pi)T}(1-t_g) + t_g)/T - \pi)$$

where  $s_d$  is the share of the real return paid out as dividends,  $r_f$  in this equation is the individual statutory tax rate, T is the capital gains holding period, and  $r_g$  is the capital gains tax rate. The first part of this equation is the after-tax return on dividends; the second part is the after-tax return on capital gains, derived from a formula of the same functional form as (9).

The tax rate on owner occupied housing is calculated by determining a cost of capital which is:

(A6) 
$$r_p = f(i(1-nt_f)-\pi) + (1-f)E(1-v) - nr_g$$

where n is the fraction of interest and property taxes deducted by homeowners and g is property taxes as a percent of asset value. The effective tax rate is increased because of the inability to deduct interest payments in full and lowered by the ability to deduct property taxes in part. Effective tax rates

levied on pre-tax income, t's, are measured as  $(r_p-r)/r_p$ ; the tax wedge,  $\tau$ , which is measured as a tax on after-tax income, is  $(r_p-r)/r$ .

The depreciation rates for equipment and structures are taken from Hulten and Wykoff (1981). Rates for rental housing and owner occupied housing are set at one percent. The inflation rate is set at .0456, the nominal interest rate is set at .0804, and the real return to corporate equities before personal tax is set at .0883. These values are those used by Hendershott and Hu (1981). The holding period for inventories is set at four months, based on the ratio of inventories to sales, and half of inventories are assumed to be covered by FIFO. The holding period for capital gains, following Bailey (1969), is set at 40 years. Data on the average holding period of corporate stock indicate a holding period of 7 years. A substantial fraction of gains, however, are held until death (most estimates suggest about three quarters) and assignment of a long holding period accounts for the failure to tax gains passed on at death. Based on data from tax returns, about half of property taxes and interest on owner-occupied housing are deducted; property taxes are estimated to be 1.4 percent of asset value. Following Fullerton, Gillette, and Mackie (1987), the share deducted falls to 40 percent after the Tax Reform Act.

There are a number of unsettled issues in the economics literature, especially as to how corporations choose debt/equity ratios and dividend payout ratios. The conventions adopted in this study use an averaging approach; alternative conventions could alter tax rates. The value of f is set at one third (see Fullerton and Henderson, 1987). Based on historical averages, two thirds of the real return on corporate equities is paid out as dividends. When aggregating capital to construct tax rates, capital stock shares are weighted by pre-tax returns. The capital stock shares reported in Table Al

are based on cumulating historical investment over time and applying depreciation weights to obtain capital stocks for 28 different business assets. These assets are allocated to industry using the capital flows tables. For further detail on the construction of assets as well as the measurement of tax lives and investments credits see Gravelle (1982, 1983). The stock of land and allocations of land among industries are taken from Eisner (1980). Allocations of inventories are taken from the Internal Revenue Service Statistics of Income. Corporate Tax Returns. 1984.

The corporate tax rate is 46 percent under prior law and 34 percent under the new tax law. The average marginal tax rates under prior and new law vary by type of income sources. In this analysis, a composite tax rate is used and is set at 30 percent under prior law and 23 percent under the new tax law, based on data from the Office of Tax Analysis, U. S. Treasury. Effective tax rates prior to the 1986 Tax Reform law are denoted by an asterisk.

Table Al: Data Used to Calibrate the Model

	Q <sub>ci</sub> /Q <sub>i</sub>	β	s	α	t <sub>c</sub> *	t <sub>n</sub> *	δ
Rental Housing	.097	.770	.05	.135	. 47	. 23	.14
Agriculture	. 293	. 580	.03	.100	. 52	. 28	. 32
Oil and Gas	.848	.776	.03	.185	. 34	.11	1.17
Mining	.844	.435	.01	.425	.37	. 20	1.54
Construction	.731	.321	. 05	.410	. 38	.09	2.31
Transportation	.923	.351	. 04	.467	. 34	. 11	1.64
Trade	.829	.421	. 20	. 386	. 52	.28	. 73
Services	,666	. 296	. 22	.315	. 41	.22	1.54
Manufacturing	1.000	.301	. 24	. 369	. 49		1.00
Utilities	1.000	.621	.06	.220	. 37		. 94
Owner-Occupied Housing	. 000	1.000	. 07	.000		. 04	. 14

Table Al continued: Data Used to Calibrate Asset Shares

	ω	ξ	ψ	89	t <sub>c</sub>	tn
Rental Housing	.00	.00	1.00		. 40	. 20
Agriculture	.10	.11	. 79	2.39	.44	.23
Oil and Gas	. 05	.09	.86	2.87	. 33	.13
Mining	.03	. 37	.60	2.93	.42	. 20
Construction	.05	.68	. 27	3.30	. 44	. 23
Transportation	. 02	. 43	. 53	2.41	.41	.19
Trade	. 36	. 14	. 45	3.87	.47	. 26
Services	.03	.42	.55	3.17	.44	. 23
Manufacturing	.30	. 31	. 39	2.64	.46	
Utilities	.04	. 24	.72	2.28	.40	
Owner-Occupied Housing	.00	.00	1.00			.04